Author's response to reviews

Title: A nationwide population-based study of the inflammatory bowel diseases between 1998 and 2008 in Taiwan

Authors:

Shu-Chen Wei (shuchenwei@ntu.edu.tw)
Meng-Hung Lin (u9656852@cmu.edu.tw)
Chien-Chih Tung (cnicemike@yahoo.com.tw)
Meng-Tzu Weng (wengmenqtiuz1@yahoo.com.tw)
Jen-Shin Kuo (internal@ms67.hinet.net)
Ming-Jium Shieh (soloman@ntu.edu.tw)
Cheng-Yi Wang (cywang@ntu.edu.tw)
Wen-Chao Ho (wcho@mail.cmu.edu.tw)
Jau-Min Wong (jmwong@ntu.edu.tw)
Pau-Chung Chen (pchen@ntu.edu.tw)

Version: 3 Date: 26 November 2013

Author's response to reviews: see over
Dear Editor,

The manuscript has been edited by a native-English speaker with scientific expertise. We attached his letter as a proof.

Sincerely,

Jau Ming Wong